Author’s response to reviews

Title: Nutritional Status of Children 7-36 Months old from Millet Consuming Communities of Masindi District, Western Uganda

Authors:

EVYLINE ISINGOMA (bisingoma@gmail.com)
Samuel Mbugua (smmbugwa@yahoo.com)
Edward Karuri (egkaruri@gmail.com)

Version: 1 Date: 28 Aug 2018

Author’s response to reviews:

The manuscript has been modified and the following additional information has been put.

1. Number of participants who completed the various assessments
2. Presentation of results has been made specific by presenting actual percentages
3. Correlation between Nutritional status of children and amounts of porridges consumed has been done
4. One additional table on dietary practices of children has been introduced
5. Mid Upper Arm Circumference results have been introduced as suggested
6. Language editing has been done throughout the manuscript
7. References have been updated
8. Clarification has been given on some aspects as requested e.g differences in sample size
9. Addition narrative has been given in the conclusion.
10. Additional information has been given on how long the study was, proportion of children that consumes millet porridges in Bujenje County of Western Uganda
11. Additional discussion of results has been made for the new aspects introduced.
12. More information is provided on the categorisation of Body Mass Index for mothers